Author's response to reviews

Title: Femoral nerve-block Intervention in Neck Of Femur fracture study - FINOF Trial

Authors:

Opinder Sahota Dr (Opinder.sahota@nuh.nhs.uk)
Iain Moppett Dr (Iain.Moppett@nottingham.ac.uk)
Nigel Bedforth Dr (Nigel.bedforth@nuh.nhs.uk)
Sarah Armstrong Dr (Sarah.Armstrong@nottingham.ac.uk)
Jim Bradley Dr (Jim.Bradley@nuh.nhs.uk)
Gerrie VanDeWalt Dr (gerrievandewelt@nuh.nhs.uk)
Martin Rowlands Dr (mrowlands1@live.co.uk)

Version: 2
Date: 4 April 2014

Author's response to reviews: see over
Dear Trials,

Many thanks to Dr Albrecht for his helpful comments with regard to our manuscript.

We have made the following changes.

Response to reviewer’s report.

Manuscript has been extensively re-edited in light of reviewer comments.

Title: Femoral nerve-block Intervention in Neck Of Femur fracture study – FINOF Trial

Version: 1 Date: 13 February 2014

Reviewer: Eric Albrecht

Reviewer’s report:

In summary, this protocol uses different synonyms and should be more consistent. The background and the discussion were probably quickly written with a lot of information presented without common thread. Outcomes should be defined. The draft should be properly edited.

Background and discussion have been reviewed and altered. Abbreviations have been removed, and meanings simplified.

Level of interest: An article of importance in its field

Quality of written English: Not suitable for publication unless extensively edited

Statistical review: Yes, but I do not feel adequately qualified to assess the statistics.

Declaration of competing interests:

No competing interest.
The manuscript has been edited to comply with guidelines for authors and the references have been inserted by EndNote with the Biomedcentral style download.

The various changes made have been tracked to allow you to follow them.

Yours sincerely,

Dr Martin Rowlands.